Author's response to reviews

Title: The Oslo Study of Clonidine in Elderly Patients with Delirium; LUCID. The protocol of LUCID, a randomised placebo-controlled trial

Authors:

- Bjørn Erik Neerland (bjorn.erik@neerland.net)
- Karen Roksund Hov (karenroksundhov@gmail.com)
- Vegard Bruun Wyller (brwylle@online.no)
- Eirik Qvigstad (eirik.qvigstad@medisin.uio.no)
- Eva Skovlund (eva.skovlund@farmasi.uio.no)
- Alasdair MJ MacLullich (a.maclullich@ed.ac.uk)
- Torgeir Bruun Wyller (t.b.wyller@medisin.uio.no)

Version: 3
Date: 8 January 2015

Author's response to reviews: see over
Cover letter

Oslo, 08 January 2015

Point-by-point response to reviewer and editor’s comments to the manuscript (version 2)

"The Oslo Study of Clonidine in Elderly patients with Delirium; LUCID. The protocol of LUCID"

Thank you once again for reviewing our manuscript and our comments. We will respond point-by-point:

Reviewer 3, B Liptzin

We don’t have any further comments to the reviewer’s report

Reviewer 2; E Eeles

We don’t have any further comments to the reviewer’s report

Editorial requirements

We have included a completed CONSORT checklist as an additional file.

We have included a CONSORT Study flow diagram as a figure (Figure 1) in our manuscript.

Revised manuscript 080115

Changes in the manuscript are highlighted in yellow. These minor changes are made to adhere to the CONSORT checklist:

Page 1: Title: We have added: “(...) a randomised placebo-controlled trial”

Page 9: Figure 1 (CONSORT Study flow diagram)

Page 10 and 11: We have specified that delirium diagnostics and inclusion procedures are made by the study physicians

Page 12: We have underlined that the capsules containing active medication and placebo will look identical

Page 19: We have added: “A full trial protocol can be accessed by contacting the corresponding author”

Page 33: FIGURE LEGENDS. Figure 1; CONSORT Study flow diagram